Author’s response to reviews

Title: ENVIRONMENTAL INJUSTICE AND CHILDHOOD LEAD EXPOSURE IN PERIURBAN (GER) AREAS OF DARKHAN AND ERDENET, MONGOLIA

Authors:

Erdenechimeg Erdenebayer (erdenechimeg.e@mnums.edu.mn)
Keilah Dos Santos (dossanke@umich.edu)
Alexjandria Edwards (alexjedw@umich.edu)
Nyam-Osor Dugersuren (nyamosor@mnums.edu.mn)
Chimedsuren Ochir (chimedsuren@hsum-ac.mn)
Jerome Nriagu (jnriagu@umich.edu)

Version: 3 Date: 20 Jan 2019

Author’s response to reviews:

RESPONSE TO EDITOR’S AND REVIEWERS’ COMMENTS

Title page

• Please include the email addresses of all authors on the title page of your manuscript. The corresponding author should still be indicated.

RESPONSE: Done as recommended

• Section headings

As indicated in our submission guidelines, https://bmcpublichealth.biomedcentral.com/submission-guidelines, please include the following sections in your manuscript: Background, Methods, Results, Discussion, and Conclusion.

RESPONSE: The titles of the sections have been amended to correspond with standard format.

• Table 3
Please correct the asterisk in Table 3, row 1, which has no accompanying footnote.

RESPONSE: Done

- Clean manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures and additional files should remain uploaded as separate files.

RESPONSE: The manuscript does not have any figures. Only the clean copy of the revised manuscript is re-submitted.